Author’s response to reviews

Title: Cognitive predictors of treatment outcome for exposure therapy: Do changes in self-efficacy, self-focused attention, and estimated social costs predict symptom improvement in social anxiety disorder?

Authors:
Isabel Kampmann (i.kampmann@uni-muenster.de)
Paul Emmelkamp (p.m.g.emmelkamp@uva.nl)
Nexhmedin Morina (morina@uni-muenster.de)

Version: 3 Date: 07 Feb 2019

Author’s response to reviews:

Dear Mrs Wollmuth,

I uploaded a single version including the tables, renamed the Methods section and mentioned the initials of all authors in the authors' contribution section.

Kind regards,

Isabel Kampmann